Author's response to reviews

Title: Relative validation of the KiGGS Food Frequency Questionnaire among adolescents in Germany

Authors:

Julia Truthmann (Truthmann@mcr.wi.tum.de)
Gert BM Mensink (MensinkG@rki.de)
Almut Richter (RichterA@rki.de)

Version: 2 Date: 12 August 2011

Author's response to reviews:

Title:
Relative validation of the KiGGS Food Frequency Questionnaire among adolescents in Germany